Author's response to reviews

**Title:** Initial experience of transumbilical laparoendoscopic single-site surgery of partial adrenalectomy in patient with aldosterone-producing adenoma

**Authors:**

Kazuyuki Yuge (keio_yuge@yahoo.co.jp)
Akira Miyajima (akiram@sc.itc.keio.ac.jp)
Masanori Hasegawa (hasem@sky.plala.or.jp)
Yasumasa Miyazaki (yasumasa0621@hotmail.com)
Takahiro Maeda (takahiro6025@yahoo.co.jp)
Toshikazu Takeda (ttakeda8156@yahoo.co.jp)
Ayano Takeda (keio-0520@wm.pdx.ne.jp)
Kazuki Miyashita (keio-1281@wm.pdx.ne.jp)
Isao Kurihara (keio-0510@wm.pdx.ne.jp)
Hirotaka Shibata (keio-0549@wm.pdx.ne.jp)
Eiji Kikuchi (eiji-k@kb3.so-net.ne.jp)
Mototsugu Oya (moto-oya@sc.itc.keio.ac.jp)

**Version:** 2  **Date:** 6 November 2010

**Author's response to reviews:** see over
Dr. Melissa Norton  
Editor-in-Chief,  
The BMC Series Editorial Production Team:  
BMC Urology

Dear Editor

We were pleased to hear the acceptance of our manuscript. We appreciate for your great assistance.
We carefully read our manuscript again and corrected the errors as you mentioned. All changes have been underlined and marked in red.  
If you have some questions, please give us email.

Sincerely yours,

Corresponding author: Akira Miyajima, M.D.  
Department of Urology, Keio University School of Medicine,  
35 Shinanomachi, Shinjuku-ku, Tokyo 160-8582, Japan.  
Phone: 81-3-5363-3824; Fax: 81-3-3225-1985;  
Email: akiram@sc.itc.keio.ac.jp